



Article Research on Supply Chain Decisions for Production Waste Recovery and Reuse Based on a Recycler Focus

Xingyao Liu ¹, Kai Wang ¹ and Hui Lu ^{2,3,*}



- ² School of Management, Fudan University, Shanghai 200433, China
- ³ Management School, Shanghai Jianqiao University, Shanghai 201306, China

Correspondence: luhui8778@yahoo.com

Abstract: Production waste recovery has economic and environmental benefits but carbon quotas limit it. To avoid future high-cost recovery technologies, we present an example of Starkelberg model between a recycler and a manufacturer, focusing on recycling exhaust gases containing metal elements from production waste. From the perspective of a recycler, this paper outlines the process of waste recycling, processing, and sales, highlighting how the proportion of recovered waste gas directly or indirectly affects sales volume. The study shows that the impact of different factors on sales volume is bifurcated, with transaction factors affecting both ordinary and new products negatively, while technical aspects positively impact new products. Surprisingly, manufacturers and recyclers benefit, even if the reasons for selling ordinary and new products are different. In the end, the products of ordinary and new in the market are mutual constraints and mutual influence.

Keywords: production waste recovery; recovered exhaust gas; remanufactured products; recycle profits

1. Introduction

The drive towards low-carbon solutions has heightened the focus on carbon constraints, including carbon quota and carbon trading. The European Union, being at the forefront of this movement, has updated its Waste Framework Directive to allow companies to recycle more in-demand recycled textiles used in the textile and garment industry [1]. Simultaneously, the Environmental Protection Bureau in China has found that the waste gases discharged from the production waste are also harmful to the environment [2]. It can thus be seen that production waste recovery can have economic and environmental benefits.

Carbon trading is also involved in the utilization of production waste. To promote sustainable practices, many countries have implemented laws and regulations related to carbon trading [3]. However, these laws and regulations can be seen as a temporary solution that only treats the symptoms, not the root cause [4]. The reason is that carbon constraints constrain market behavior and have adverse effects during production waste.

On the one hand, carbon constraints constrain the market behavior of enterprises' voluntary trading. They noted that the market behavior is not spontaneous but due to the policy supervision penalty. For example, according to the Ministry of Ecology and Environment, the total value of carbon emission quotas reached 179 million tons, with a policy supervision penalty of 7.661 billion yuan and a 99.5% completion rate of performance [5]. On the other hand, the carbon quota for production waste has not been fully achieved. Germany, for instance, the world leader in recycling, only recycles 16% of composite products [6]. The complex technology, separation, and processing required to recycle production waste often result in only a small part of it being reused.

It is an excellent opportunity to propose that enterprises themselves deal with production waste recovery and reuse through another market mechanism. Suzhou Jinhong Gas Co., Ltd., a leader in China's solid waste industry, is the second largest after the new



Citation: Liu, X.; Wang, K.; Lu, H. Research on Supply Chain Decisions for Production Waste Recovery and Reuse Based on a Recycler Focus. *Sustainability* **2023**, *15*, 3320. https:// doi.org/10.3390/su15043320

Academic Editors: Mei-Fang Chien and Tanmoy Roy Tusher

Received: 13 October 2022 Revised: 28 January 2023 Accepted: 7 February 2023 Published: 10 February 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). energy industry [7] and has become one of the few monopolies in China that deals with production waste. This company has the advantage of being a one-stop shop for recovery, production, and purification, making it a fully integrated gas provider. However, this is a challenging task for most ordinary enterprises, as it requires a significant investment and the profits are limited [8]. Even Jinhong regards production waste recovery and reuse as a unique business rather than its main business [9].

In addition, such recovery is mostly energy recovery after incineration rather than reusing production waste. Most of the recovery is incinerated or landfilled in the United States and Britain, i.e., Lidl in Germany and Mura in Britain [10]. Lidl is one of Germany's most considerable waste recycling retail giants by controlling the overall energy recycling cycle through the production and sales of goods [11]. Mura is finding a third-party partner to recycle the plastic that has been incinerated and buried. Both sides have fulfilled their commitment to environmental sustainability so far [12]. It can be seen that most situations cannot recycle production waste, and the reuse of production waste is also rare.

Most research is related to the process of end-of-life product recovery, treatment, and reuse, which reference consumers' low-carbon preference and the recycling conditions. It should be noted that consumer preference should consider the price difference between new and remanufactured products [13]. Sometimes the price of low-carbon new products and ordinary waste may be the opposite [14]. Additionally, product recovery value and price conditions can affect the decision-maker's choice of social challenges [15]. Therefore, by improving the recovery conditions, the rate of production waste recovery can be increased, bringing economic and environmental and social benefits to the entire system.

Above all, this paper considers that enterprises deal with production waste through a new market mechanism. The relevant regulations on carbon constraint now specifically require enterprises for environmental–social–economic sustainable development to pay attention to such issues again [16]. Therefore, production waste recovery and reuse should carefully consider the recyclers' behavior in the market mechanism when analyzing the carbon quota situation. The reason is that the manufacturer can supply the production waste and process ordinary products.

Moreover, recyclers can sell new products to the market, competing with the ordinary products sold by the manufacturer. The recycler acts as a dealer simultaneously and has a transaction relationship with the producer. Other vital points we explore in this paper are the following:

- (1) How can the manufacturers guarantee a profit increase when supplying the exhaust gas to recyclers?
- (2) How do recyclers take the best decision in the process of production waste recovery and reuse with or without a carbon quota?
- (3) Should the government participate in production waste recovery and the reuse process to make the scheme more economical and effective?
- (4) What is the market impact of different factors on the decision-making of ordinary and new products?

It should be noted that the recycler is usually a third party and is affected by consumers' preferences, competition, cost, and other factors that influence consumers' sales decisions and profits [17]. Therefore, recyclers must consider a different price strategy when studying the coordination mechanism to achieve their best profit [18]. For example, Suzhou Moore Gas Co., Ltd. recovers nitrogen from various trades, including electronics, the chemical industry, optical fiber, and metallurgy, to achieve the best profit [19]. Because manufacturers choose different channels, their wholesale prices and retailers' marketing effects will be distinguished [20]. Therefore, this paper uses the recycler as the third party when building its model.

The result was beyond our expectations when the proportion of waste gas recovery was proposed for recovering gas containing metal elements from the production waste process. This study introduces a novel approach to connect the traditional supply chain to the gas supply chain, providing valuable insights into the management field and helping enterprises fully grasp the benefits of product quality assurance. As a third party in the waste recycling and reuse process, the recycler can enhance the utilization efficiency of products and achieve the sustainable goals of the Green supply chain. More attention here is paid to the influence of specific factors in the research process than before when taking the proportion of recovered exhaust gas as a new entry point of technological innovation.

The rest of this paper is organized as follows: Section 2 reviews the relevant literature. Section 3 describes model problems and fundamental assumptions. Section 4 analyzes the optimal results without considering the carbon quota and then considering the carbon quota. Section 5 offers a comparative analysis of the above two situations. Section 6 is an example analysis. Section 7 summarizes the paper and its findings and expounds on the significance of this study for the future development of production waste recovery and its reuse.

2. Literature Review

2.1. Waste Recycling and Reuse Analysis

This paper focuses on production waste recovery and reuse. While most studies on recovery, treatment, and reuse have concentrated on ordinary or end-of-life products. [21]. It has also been discovered that there is no significant interaction between the different waste recovery rates, renewable energy, and carbon dioxide emissions in terms of product recovery rate [22], which is related to advertising efficiency and product order quantity [23].

When there is no third-party involved, producer-led independent recycling channels are the most efficiency for channel selection [24]. These channels can sometimes maximize the total profit of the supply chain through government subsidies [25]. At that time, manufacturers and recyclers will conduct price competitions and cooperation [26]. In contrast, when considering the third party, the best choice is to use the recycling platform for old goods recycling [27]. Specifically, such competition between third-party recyclers can reduce product prices and improve the recycling efficiency of waste products but also play a positive role in promoting income level [28].

Most scholars have adopted the empirical research method to consider production waste recovery, treatment, and reuse. However, the effectiveness and efficiency of municipal production waste recycling are unsatisfactory due to poor waste segregation at the generation source [29]. Using field observations, semi-structured interviews, and secondary data sources, the structural model method can better analyze the interaction in the end [30].

However, there are few theoretical models on the complete production waste recycling and reuse process. Even if there were, it would refer to unilateral recycling of production waste [31]. Moreover, theoretical models convert discarded waste into energy for utilization [32]. Analyzing the public's perception of risk is thus the best way to deal with waste-to-energy [33]. Some documents, when considering third parties, pay attention to everyone, when sending a revised version of MS to the reviewers, but we only need to highlight the modified content, without having track changes on, and without emphasizing any deleted content, and so function as a recycling platform in this way [34]. Therefore, the case suggests that production waste recycling solves the pricing and profit problem by analyzing the relationship between the recycling rate and consumers' recycling willingness [35].

2.2. An Analysis of Remanufacturing

This analysis aligns with the public's idea that remanufactured products are not as good as new ones. That recognition will lead many people to believe that remanufactured products should be "the cheaper, the better" [36]. The challenge remanufacturers face is that it can be difficult to accurately determine the condition of used products before disassembly and inspection [37]. However, research has shown that while the optimal sales price of new and remanufactured products increases, the recovery price and social welfare decrease [38]. Distribution channel choice [39] thus should be used as a potential factor to foster more remanufactured products [40] and their sales [41].

It should be noted that while manufacturers can benefit from adopting emission reduction technology, recycling, and a remanufacturing approach, they cannot guarantee the minimum total carbon emissions in the supply chain [42]. In addition, remanufacturing should consider the impact of policy because the distribution of government subsidies has an incentive effect on both manufacturers and consumers [43]. However, remanufacturers must also meet other requirements, such as obtaining patent authorization and addressing equity concerns, to turn a profit [44]. Remanufacturing balances economic interests and environmental impact more effectively than tax and carbon reduction policies [45]. Therefore, many scholars have proposed that more insight is required into consumer expectations and willingness [46]. For example, the research should focus on differential pricing and the profit distribution of new and ordinary products [47].

2.3. Analysis of the Carbon Quota

By combining transaction prices and incorporating multiple entities, the allocation of carbon quota can have varying effects. Therefore, in any allocation and use of carbon quota, a government enterprise consumer tripartite carbon quota allocation can be established [48]. It can also improve the participation enthusiasm of consumers and enterprises based on maintaining the best social welfare. Thus, both meet fairness and efficiency requirements and lower emission reduction costs [49].

Although the carbon quota policy can improve the sustainable level of inventory and transportation, it has no impact on the service level [50]. Therefore, when considering the carbon quota and its transaction price, enterprises need to pay more attention to environmental and consumers' Green preferences, such as marketing strategies related to advertising [51] and sales [52], as well as Green goodwill [53]. Indeed, given differential pricing and unified pricing under a carbon quota, enterprise profits will be affected by differences in competition and cooperation [54].

2.4. Waste Audit in the Supply Chain

The field of production waste recovery and reuse holds immense potential and value, as it remains a relatively untapped area of research. Despite this, the existing studies have primarily focused on solid waste and have not yet fully explored the implications of a carbon quota in the waste recovery and treatment process.

Supply chain audit is related to cost and inter-subject competition [55]. On the one hand, partners input costs to operate a prediction market for sharing their forecasts and resolving their differences through costly actions [56]. On the other hand, competition among different entities can harm audit quality but can favor cost pricing [57]. Even waste audits can be used as an environmental management tool to help regulators grasp information on waste management and guide enterprises to carry out waste risk management [58]. Surprisingly, when audits are tied to recycling, they can drive green and sustainable development [59].

In conclusion, there is much room for exploration in the theoretical production waste recovery and reuse model. The research on recycling and reusing production waste also focuses on an empirical research method. Still, the review of the related literature suggests that the existing studies on waste recovery and treatment have mainly focused on solid waste. Besides, the market mechanism of production waste can be attractive when considering the carbon quota because most of the research on the carbon quota has nothing to do with the recycling and reuse of production waste. This research can examine the influence of different factors, such as cost and competition, on the carbon quota. In the next section, we present a description of our modeling framework.

3. Model Preparation

3.1. Problem Description

In this paper, a supply chain model consisting of a manufacturer and a recycler is considered from the recycler's perspective, taking waste gas recovery as an example. In detail, there are two sales channels: (1) the manufacturer produces ordinary products and deals with production waste; (2) the recycler then recycles the production waste of ordinary products and makes new products for sale to the market. It's worth mentioning that recycling refers to the exhaust gas containing metal elements during waste production. On the one hand, the recycler can recover the waste gas from the production waste collected from the waste discharged from the manufacturer's ordinary products. On the other hand, recyclers convert the recovered gas into new products and sell them to the market. The model's overview is illustrated in Figure 1.



Figure 1. Model diagram for waste recycling.

Some similarities and differences should be noted as well. Like [20,47], manufacturers choose different channels to distinguish between new and ordinary products' prices. There are ordinary products that have not been recycled and new products that have been recycled and reused in the market. Under the Stackelberg game, the manufacturer is the leader, and the recycler is the follower. The difference is that the game is played between the producer and the recycler, resulting in different prices, costs, and sales volumes. It is also important to note the wholesale price.

Combined with [49,54], we must consider the following problems. First, we must consider market trading risks to guarantee the manufacturers' profit. Second, we can analyze whether recyclers need to consider carbon emissions when making optimal decisions. Next, we can explore whether this model result is affected by different factors. Last but not least, we need to consider which scheme is more economical in the situation of competition and cooperation for the government. Therefore, enterprises need to consider cost, sales volume, profit, recovery ratio, and other factors in detail.

3.2. Basic Assumptions

The fundamental assumption is that the recycler recovers the waste gas containing metal elements to make new products. In detail, if a unit of ordinary product can emit m tons of waste gas, the total emission of exhaust gas is $m \cdot q_s$ when the carbon quota is not considered and are $\beta \cdot m \cdot q_s$ when the carbon quota is considered. Taking the former equation as an example, if the proportion of waste gas recovered by the recycler is $\tau(0 < \tau < 1)$, then the amount of waste gas recovered is $\tau \cdot m \cdot q_s$; if the gas concentration ratio is $\delta(0 < \delta < 1)$, then the amount of gas containing metal elements is $\delta \cdot \tau \cdot m \cdot q_s$; if the ratio between the gas containing metal elements and the new product is σ , then the number of new products sold by the recycler is $q_r = \frac{\delta \cdot \tau \cdot m}{\sigma} q_s$ (Table 1). Therefore, it can be inferred that:

- (1) the profit function for the recycler is $\pi_r = (p_r c_r) \cdot \frac{\delta \cdot \tau \cdot m}{\sigma} q_s c_t \cdot \delta \cdot m \cdot \tau \cdot q_s w \cdot \tau \cdot m \cdot q_s$,
- (2) the profit function of the manufacturer is $\pi_m = w \cdot \tau \cdot m \cdot q_s + (p_s c_s) \cdot q_s$,
- (3) the demand function of new products is $p_r = \theta \cdot [Q q_s(\frac{\delta \cdot \tau \cdot m}{\sigma} + 1)]$,

Symbol	Illustration
Q	Potential market demand
т	The quantity of exhaust gas that can be generated per unit of ordinary products $(m > 0)$
β	The proportion of allowable emissions affected by the carbon quota $(eta \in (0,1])$
w	The production cost of wholesale waste from the manufacturer to the recycler
τ	The proportion of recovered exhaust gas
δ	Gas concentration ratio
σ	Gas conversion ratio
Cs	The production cost for the manufacturer selling ordinary products to the market $(c_s > c_r/c_t)$
Ct	The production cost for recycler refining gas containing metal elements
Cr	The production cost of new products made by recyclers
p_s	The unit price for ordinary products the manufacturer sells to the market
pr	The unit price at which the recycler sells new products to the market $(p_r > p_s > w)$
q_s	The number of ordinary products sold by the manufacturer
q _r	The number of new products sold by the recycler $(q_r \ge q_s \ge 0)$

Table 1. Parameters and Symbols of Production Waste Process.

When considering the carbon quota situation and so on, this study does not elaborate. Other basic assumptions for the model notions are as follows:

- (1) All new products will have the same quality and can be remanufactured and sold.
- (2) To simplify the model expression, let Q = 1, $\frac{\delta}{\sigma} = 1$, m = 1. When the carbon quota is not considered, $q_r = m \cdot q_s$ when the carbon quota is considered,

$$q_r = \beta \cdot \tau \cdot q_s \tag{1}$$

(3) Consumers' willingness to buy ordinary and new products may differ. In other words, consumers' acceptance of ordinary and new products has a consumer psychological preference θ , and $0 < \theta < 1$. Currently, the demand function is

$$p_r = \theta \cdot [1 - q_s(\tau + 1)] \tag{2}$$

$$p_s = 1 - q_s(\theta \cdot \tau + 1) \tag{3}$$

4. Remanufacturing Model Results

4.1. Without Considering the Carbon Quota

It should be noted that the influencing factors in this paper can be roughly divided into two categories in terms of production waste recycling, treatment, and reuse: technical factors and transaction factors. In particular, the technical factors are mainly the carbon quota ratio, gas concentration (conversion) ratio, and recycled waste gas ratio that need attention for the process technology. However, transaction factors are price and cost in sales businesses.

In addition, the waste gas has gone through recovery, refining, and sales in a single cycle. When the price of wholesale gas is w, the sales price of the products sold by the manufacturer to the market and the price of the recycler are p_s and p_r , respectively. If

necessary, a distinction needs to be made between general and optimal cases of factors, such as sales volume and price.

Based on the sales behaviors of a manufacturer and a recycler, the formulas for the manufacturer to sell ordinary products and the recycler to sell new products are as follows: The profit function for the manufacturer is:

$$\pi_m = w \cdot \tau \cdot q_s + (p_s - c_s)q_s \tag{4}$$

The profit function for the recycler is:

$$\pi_{r1} = (p_r - c_r - \delta \cdot c_r - w) \cdot \tau \cdot q_s \tag{5}$$

Proposition 1. The cost of gas-containing metal elements in the recovery, treatment, and reuse of production waste has a negative impact on producers but a positive impact on recyclers.

See Appendix A for the proposition text and inference process in this paper.

The cost of containing metal elements is the initial price of refining metal-containing gases. Its impact on different channels reflects the changes in manufacturers and recyclers in some respects. Specifically, the adverse effects of gas-containing metal elements on ordinary products are reflected in the price of wholesale gas and the sales volume of ordinary products. In contrast, the positive impact on new products is reflected in the sales volume of those new products in Corollary 1 and Corollary 2 below.

Corollary 1. Without considering the carbon quota, the optimal price of a producer's wholesale gas decreases with an increase in the cost of the recycler's refining of the gas-containing metal elements.

Corollary 2. Without considering the carbon quota, the optimal sales volume of ordinary products decreases with an increase in the cost of recovering the gas-containing metal elements by the recycler, while the optimal sales volume of new products increases with an increase in the cost for retrieving the gas-containing metal elements.

The results from Corollarys 1 and 2 reveal that cost is one of the primary transaction factors affecting the optimal sales volume of ordinary products. The increase in the cost of gas-containing metal elements is due to the poor quality control measures for waste recovery products, which increases the scrap cost [60]. However, with the support from the government for the growth of the environmental protection sector, the optimal wholesale gas cost still tends to decrease, even with rising scrap costs [61]. This prompts recyclers to purchase more waste gas at a lower wholesale cost and sell new products with a small profit margin, leading to an increase in the sales volume of new products is reduced by the competition with similar products in the market. This reinforces the notion that supply chain waste audits are related to cost, and competition and that waste recovery and reuse are impacted by these factors.

Proposition 2. Ordinary products are negatively affected by transaction factors, such as cost and price, in the process of production waste recovery, treatment, and reuse.

Here, only the transaction factors affecting the general sales volume and optimal sales volume of ordinary products are explained in Corollary 3. More specifically, the general cost of wholesale gas negatively affects the optimal sales of ordinary products. New products are analyzed in Section 6.2, which mainly discusses technical factors. Furthermore, Corollary 4 summarizes the production cost range of new products, which is generated and extended under the mutual relationship between wholesale pricing and the sales volume of ordinary products.

Corollary 3. Without considering the carbon quota, the optimal sales volume of ordinary products will decrease with an increase in the general cost of wholesale gas.

Corollary 3 proves that the optimal sales volume of ordinary products and the general cost of wholesale gas change inversely. The details are as follows: when the optimal sales volume is low, the general cost of wholesale harms the optimal sales volume of ordinary products. The reason is that the larger the wholesale price, the more substantial the profit the manufacturer receives from the recycler, and the less the investment in the ordinary products, resulting in a decrease in the optimal sales volume of the ordinary products.

Further, combining Corollary 1 with 2, it is found that the optimal sales volume of ordinary products and the optimal cost of wholesale gas show positive changes. Still, the optimal sales volume of ordinary products and the general cost of wholesale gas show negative changes. It can be seen that cost stability affects the optimal sales volume of ordinary products. In contrast, the wholesale gas cost strongly fluctuates the optimal sales volume of ordinary products.

Corollary 4. When the carbon quota is not considered, and $\theta \in (0, 1)$, $c_s \in (0, 1)$, $\delta \in (0, 1)$;

- (1) *if* $\frac{\theta c_t}{\delta} < 0$ and $\frac{\theta c_t}{\delta} > 1$, and $2c_s \delta \cdot c_t + \theta 2 \in (0, 1)$, $c_r \in (0, 2c_s \delta \cdot c_t + \theta 2)$;
- (2) if $0 < 2c_s \delta \cdot c_t + \theta 2 < 1 < \frac{\theta c_t}{\delta}$, $c_r \in (0, 2c_s \delta \cdot c_t + \theta 2)$;
- (3) if $0 < \frac{\theta c_t}{\delta} < 2c_s \delta \cdot c_t + \theta 2 < 1, c_r \in \emptyset$;
- (4) if $0 < 2c_s \delta \cdot c_t + \theta 2 < \frac{\theta c_t}{\delta} < 1$, $c_r \in (2c_s \delta \cdot c_t + \theta 2, \frac{\theta c_t}{\delta})$; (5) if $2c_s \delta \cdot c_t + \theta 2 < 0$ and $\frac{\theta c_t}{\delta} > 1$, $c_r \in (0, 1)$.

Proposition 3. The proportion of waste gas recovered plays an intermediary role in the process of production waste recovery, treatment, and reuse. It has a positive impact on both new products and ordinary products.

The proportion of recycled waste gas plays a crucial role in the relationship between the cost and sales volume of new and ordinary products. This proportion is a dynamic factor that is affected by and affects other factors in the waste recovery process. In other words, the proportion of waste gas recovered is affected by the cost and the sales volume of new and ordinary products, which is beneficial. The impact on the proportion of waste gas is explained in Corollary 5, while the reason it affects ordinary products and new products is described in Corollary 6. Additionally, the policy and the production cost of new products affect the proportion of recycled waste gas, as illustrated in Corollary 10 and Section 6.2.

Corollary 5. Without considering the carbon quota, the optimal proportion of the recovered exhaust gas increases with the increased cost the recycler pays to extract the gas-containing metal elements.

Corollary 6. Without considering the carbon quota, the optimal sales volume of ordinary products and the general sales volume of new products will increase with an increased general proportion of recovered exhaust gas.

Corollary 6 shows that when the optimal proportion is not reached, the proportion of recovered exhaust gas positively impacts the sales volume of both ordinary and new products. Specifically, in terms of technology, the higher the ratio of waste gas recovery, the greater the augment in the amount of waste gas emitted. Because the production of ordinary products emits exhaust gas, the number of ordinary products is increasing. As a result, the increased production of ordinary products leads to a higher sales volume. All in all, different from the wholesale price of gas in Corollary 3, the proportion of recycled waste gas is the second reason that will affect the optimal sales volume of ordinary products. The optimal sales volume of new products is affected not only by the general proportion of waste gas recovered but also by other factors, as detailed in Section 6.2.

Proposition 4. *Remanufactured and ordinary products are mutually restricted and occupy the market in the competition.*

The production cost of ordinary products is a key transaction factor that can significantly impact the market. It is not as common for the price of competitive products to decline even if the cost of ordinary products increases. However, the increase in the price of ordinary products causes the pitch in their prices, which is very rare for the price of competitive products to decrease. Proposition 4 indicates that if ordinary products change in a certain respect, new products will also be affected. Moreover, the market will show a declining situation and another rising. The specific description is described in Corollary 7.

Corollary 7. Without considering the carbon quota, the optimal sales volume and optimal price of ordinary products will increase with the production cost of ordinary products. In addition, the optimal sales volume and optimal price of new products will decrease with the increase in the production cost of ordinary products.

From Corollary 7, the production cost of ordinary products has a positive impact on the optimal price of ordinary products and a negative effect on the optimal price of new products. As with the price transmission and transfer effect of Jinfei Kaida company, the sales volume is the inventory, and the sales price is the sales value [62]. Known sales price = $\cot x$ sales volume + other. In other words, sales value = $\cot x$ inventory + additional [63]. When the production cost of ordinary products rises, then unsalable ordinary products will increase the inventory, increasing the value of the accumulated sales products. In contrast, when the production cost of ordinary products rises, the inventory of new products decreases due to the substitution of new products, resulting in a decrease in the accumulated value of the new products.

4.2. Considering the Carbon Quota

The carbon quota limits the carbon emission amount for factories and enterprises. In essence, the carbon quota limits the total amount of waste gas emitted by manufacturers and thus affects recyclers' recycling, refining, and sales processes. Therefore, the channel for manufacturers to sell ordinary products is unchanged, and the profit function of the recycler is as follows:

$$\pi_{r2} = (p_r - c_r - \delta \cdot c_r - w) \cdot \beta \cdot \tau \cdot q_s \tag{6}$$

Proposition 5. The environmental situation that considers the carbon quota has a significant impact on recyclers, and policy is one of the factors affecting the proportion of waste gas recovered.

There are some similarities and differences between not considering a carbon quota and considering a carbon quota. Corollary 8 shows that manufacturers have more similar results for changes in different environments. Different parts are primarily from recyclers, which are explored in greater detail in the case study section. It is worth noting that the policy factor is neither a technical factor nor a transaction factor. The influence of policy on the proportion of recovered waste gas is mandatory, as demonstrated in Corollary 10. Additionally, as outlined in Corollary 9, the cost is a limiting factor, but it does not violate conventional wisdom.

5. Comparison and Analysis

The results from this study demonstrate the optimal solutions for the two cases, considering and not considering the carbon quota, allowing for decision-making.

5.1. Comparison and Analysis of Optimal Values

Corollary 8. The optimal solution satisfies the conditions without considering the carbon quota and the carbon quota (1) $w_1^* = w_2^*$, $q_{s1}^* = q_{s2}^*$; (2) Corollary 1, Corollary 2, Corollary 3, Corollary 6, and Corollary 7.

Corollary 8 demonstrates that when analyzed with the aforementioned findings, the similarities between not considering the carbon quota and the carbon quota mainly relate to the content of trend change, and most similarities are related to the producers. The trend changes in (2) have one side increasing, resulting in the other side also changing. However, as opposed to Corollary 5, it is revealed that the impact of different scenarios has a more pronounced effect on recyclers. In essence, different situations will have a more substantial effect on recyclers than on manufacturers.

Corollary 9. (1) If $\delta \in (0, 1)$, $c_r > c_t$ and $c_r/c_t \in (0, 1)$, $p_{s1}^* < p_{s2}^*$, $p_{r1}^* < p_{r2}^*$; (2) If $\theta \in (0, 1)$, $c_r \neq c_t$, and $c_r/c_t \in (0, 1)$, $\pi_{m2}^* > \pi_{m1}^*$.

Corollary 9 illustrates the point that cost is also a constraint. There may be other costs, but the results can only be suitable under limited conditions. Combined with the above assumptions, we will obtain: $c_s > c_r > c_t$. The relationship between price and profit under certain conditions is also shown here, paving the way for the following analysis.

5.2. Trend Comparison and Analysis

Corollary 10. The difference between considering the carbon quota and not considering the carbon quota is that the optimal proportion of the recovered exhaust gas decreases with an increase in the cost for the recycler to extract the gas containing metal elements.

The results of this model comparison demonstrate the impact of considering or not considering the carbon quota on the proportion of recovered waste gas. According to Corollary 5, the increase in the proportion of recovered waste gas without considering the carbon quota results from the low cost of wholesale gas purchases. On the other hand, reducing the proportion of recovered waste gas when considering the carbon quota is due to the direct limit on the amount of waste gas discharge. Therefore, increasing the cost of refining metal element gas cannot change policy implementation.

6. Example Analysis

Some relationships cannot be directly derived by analyzing value from the model process, such as the optimal proportion of recycled exhaust gas, the sales volume, the sales price of ordinary and new products, and the relationship between the profits of recyclers. Therefore, this paper uses the numerical simulation method to explore the above conclusions and obtain more laws for waste recycling. After referring to the internal data of several enterprises to understand the relevant situation, different parameters were set as follows: $c_s = 0.9$, $\theta = 0.5$, $\delta = 0.5$, $c_t = 0.1$. According to the above analysis and without considering the carbon quota, and considering the carbon quota, $c_r \in (0.25, 0.4)$.

6.1. Comparing and Analyzing the Optimal Proportion of Recovered Waste Gas

According to the parameters $\tau_1^* = \frac{-11c_r+3.5}{4c_r-1}$, $\tau_2^* = \frac{0.8-2c_r}{c_r-0.25}$, the relationship between the optimal proportion of recovered exhaust gas and without considering the carbon quota and considering the carbon quota in Figure 2. The detailed description of this image is as follows: (1) The optimal proportion of recovered exhaust gas decreases with an increase in the production cost of new products without considering the carbon quota and the carbon quota. (2) The optimal proportion of recovered exhaust gas under consideration of a carbon quota is higher than without any consideration of a carbon quota ($\tau_2^* > \tau_1^*$). Combined

with Corollary 5, the result explains that the cost of gas containing metal elements positively impacts the proportion of recovered waste gas. In contrast, new products' production cost negatively impacts the recovered waste gas ratio.



Figure 2. The new product production cost affects τ_1^* and τ_2^* .

6.2. The Optimal Sales Volume of New Products under Different Circumstances

According to the parameters $q_{r1}^* = -0.9167c_r + 0.2917$, $q_{r2}^* = -0.6667\beta(c_r - 0.40)$, $q_{r1}^* - q_{r2}^* = -0.9167c_r + 0.2917 + 0.6667\beta(c_r - 0.40)$, the relationship between the optimal sales volume of new products without considering the carbon quota and considering the carbon quota is analyzed in Figure 3. The details are as follows: (1) The optimal sales volume of new products without a carbon quota decreases with the increase of c_r , and the optimal sales volume of new products with a carbon quota decreases with the growth of c_r , along with the rise of β . (2) The optimal sales volume of new products considering a carbon quota is more than that for new products considering a carbon quota. This trend is because $q_{r1}^* - q_{r2}^*$ increases with $c_r \in (0.25, 0.4)$, and $\beta \in (0.25, 0.4)$ shows a trend of at first greater than zero and then less than zero.



Figure 3. The influence of different factors on q_{r2}^* (left) and $q_{r1}^* - q_{r2}^*$ (right).

Therefore, the cost is one of the main transaction factors that affect the optimal sales volume of ordinary products. Corollary 2 is verified. Combined with Proposition 1, the production cost of new products hurts the optimal sales volume of both ordinary and new products. Moreover, the optimal sales volume of new products is also affected by the carbon quota ratio and the concentration or conversion ratio (which can be obtained by the same process in addition to the cost). The content of the explanation responds to Corollary 6. This positive effect will eventually cause the optimal sales volume of the carbon quota to exceed the optimal sales volume without considering the carbon quota.

It can be seen, therefore, that the Green supply chain will adhere to the carbon quota and gradually improve the quality of raw materials, and finally achieve good sales results. New products are positively affected by technical factors, such as the proportion of recycled waste gas, the carbon quota ratio, and the concentration or conversion ratio, which also proves that the technology mentioned before is one of the factors that cannot be ignored. In addition, transaction factors, such as new product production costs, also hurt new products' sales.

6.3. A Comparison and Analysis of Optimal Product Prices

According to the parameters $p_{s1}^* = 0.125c_r + 0.9375$, $p_{s2}^* = 0.9$, $p_{r1}^* = 0.3958 + 0.2917c_r$, $p_{r2}^* = 0.4083 + 0.1667c_r$, the price relationship between ordinary products and new products is analyzed in Figure 4. Synthesized the results of the three maps, without considering the carbon quota and considering the carbon quota, given the increase c_r , the optimal sales price of ordinary products is higher than that of new products. That is to say $p_{s1}^* > p_{r1}^*$, $p_{s2}^* > p_{r2}^*$. In addition, since $c_r > c_t$ verifies $p_{s1}^* < p_{s2}^*$, $p_{r1}^* < p_{r2}^*$ for Corollary 7 (1), we get $p_{r1}^* < p_{s1}^* < p_{s2}^*$. Compared with p_{s1}^* and p_{r2}^* , it can be seen that $p_{s1}^* > p_{r2}^*$ within the range of c_r , so $p_{r1}^* < p_{r2}^* < p_{s1}^* < p_{s2}^*$. Therefore, the rising production cost of new products has a positive impact on the optimal price of ordinary products and a negative effect on the optimal price of new products.



Figure 4. The analysis of new product production cost on p_s and p_r .

6.4. A Comparison and Analysis of Product Optimal Profit

 $\pi_{r1}^* = 0.0556(2.75c_r - 0.875)^2$, $\pi_{r2}^* = 0.2222\beta(c_r - 0.40)^2$, $\pi_{m2}^* = 0.05500 - 0.3000c_r + 0.3333(c_r + 0.05)^2$, $\pi_{m1}^* = -1.000(-0.2500c_r + 0.2000)(-0.2917 + 0.9167c_r) + 0.3333(0.1250c_r + 0.0375)(c_r - 0.25)$, the figure thus analyzes the profit relationship between producers and recyclers and is verified by the above parameters that let Corollary 7 (2) get $\pi_{m2}^* > \pi_{m1}^*$. One more note, the color of the picture is to show that it is a 3D picture. Details are as follows:

First, compare the relationship between $\pi_{r2}^* - \pi_{m2}^*$, $\pi_{r1}^* - \pi_{r2}^*$, $\pi_{r1}^* - \pi_{m1}^*$. The details are as follows: (1) Under the consideration of a carbon quota, the overall profit of recyclers is less than that of producers with c_r and β . As shown panorama of the model in Figure 5a, the color indicates that the darker the range of influence the smaller and the lighter the range of influence the larger. (2) With the increase of c_r , the profit of recyclers without considering the carbon quota is less than that of the recyclers without considering the carbon quota first and more remarkable than that of the recyclers without considering the carbon quota. As shown in the detail diagram of the model in Figure 5b, white and color represent the difference, with white representing a difference less than zero and color representing a difference greater than zero. (3) Without considering the carbon quota, the profit of the recycler will be less than that of the producer with the increase c_r .



Figure 5. The overall profit comparison and analysis chart of π_r and π_m .

It can be seen, therefore, that the optimal profits of recyclers and producers affect each other. Considering that the carbon quota is beneficial to the overall yield of producers, not considering the carbon quota is advantageous to the optimal profit of recyclers. Of them, the rising production cost of new products is conducive to the gain of recyclers without considering the carbon quota; the increase in the proportion of compressed gas is also conducive to considering the profits of recyclers under the carbon quota.

Secondly, based on the above optimal profit, when comparing the manufacturer's internal profit and the recycler's profit, the internal profit of the manufacturer is divided into the profit from wholesale gas and the profit from ordinary products, and the profit of the recycler is then the profit from new products. These results, without considering the carbon quota and considering the carbon quota, satisfy the following requirements: (1) The optimal profit of wholesale gas is more significant than that of ordinary products first and then smaller. Because $\tau_1^* \cdot w_1^* \cdot q_{s1}^* - (p_{s1}^* - c_s)q_{s1}^*$ and $\tau_2^* \cdot w_2^* \cdot q_{s2}^* - (p_{s2}^* - c_s) \cdot q_{s2}^*$ are first greater than zero and then less than zero with the increase of c_r . (2) The optimal profit for wholesale gas is less than that for new products first and more significant than for new products. Because $\tau_1^* \cdot w_1^* \cdot q_{s1}^* - \pi_{r1}^*$ with the increase of c_r , $\tau_2^* \cdot w_2^* \cdot q_{s2}^* - \pi_{r2}^*$ with the rise of c_r and β , all are first less than zero and then greater than zero. (3) The optimal profit of ordinary products is less than that of new products. Because $(p_{s1}^* - c_s)q_{s1}^* - \pi_{r1}^*$ and $(p_{s2} - c_s)q_{s2}^* - \pi_{r2}^*$, the rise of c_r and β is first less than zero and then greater than zero. The color change represents the same meaning as Figure 5b.

Thus, when the production cost of new products is small enough, the optimal profits for wholesale gas and new products are greater than those for ordinary products, and the earnings of new products are greater than those of wholesale products. However, at this time, the producers and recyclers have optimal profits, and the market will reach a stable state (see Figures 5 and 6). Unlike reference [18], this point does not use the coordination mechanism to receive the profits.



Figure 6. The internal profit comparison and analysis chart for π_r and π_m .

7. Conclusions and Final Thoughts

7.1. Conclusions

This text explores the optimization challenge faced in the waste recycling process. It begins by analyzing the optimal price, sales volume, and profit achieved through two separate sales channels for new and ordinary products. The next step involves an examination of the influence of factors such as recovery ratio, costs, and others on the optimal results when considering or not considering carbon emissions. The final section provides a conclusion on the decision-making for the sales volume, price, and profit of both new and ordinary products through further analysis and examples. The conclusions are as follows:

- (1) From the perspective of the proportion of recovered exhaust gas, the general proportion of recovered exhaust gas will directly affect the general sales volume and the optimal sales volume. Otherwise, it also indirectly serves as the intermediary of cost and policy when it is the optimal proportion of waste gas recovery. Among these, the cost of gas-containing metal elements has a positive impact on the optimal proportion of recovered exhaust gas, the cost of new products has a negative effect on the optimal proportion of recovered exhaust gas, and the impact of national policies that are not affected by other factors is a mandatory one.
- (2) Common products and new products under optimal sales volume and optimal sales volume are affected by different reasons and degrees in product sales. Technical factors positively impact new products, while transaction factors hurt both. Specifically, the cost of gas-containing metal elements and the general cost of wholesale gas negatively affect the optimal sales volume of ordinary products. Additionally, the optimal sales volume of new products is positively affected by the proportion of carbon quota, the proportion of gas concentration (conversion), and the cost of gas-containing metal elements. Still, it is negatively affected by the production cost of new products. Besides, the proportion of the recovered waste gas positively affects the general sales volume of new products.
- (3) The decision-making for recyclers, in terms of optimal prices and profits, is primarily driven by revenue. For example, when the production cost of new products rises, it is beneficial for the optimal price of ordinary products and the optimal profit of recyclers without considering the carbon quota. When the production cost of new products drops, the yield of new products is more significant than that for ordinary products, yet both producers and recyclers benefit.
- (4) New and ordinary products are constrained by each other, resulting in mutual influence in the marketplace. Just as ordinary products sell poorly, new products sell well. Thus, new products will drive ordinary products out of the market without considering the carbon quota. On the contrary, the two will balance each other in the market when considering the carbon quota.

Overall, the cost of gas-containing metal elements positively impacts the optimal sales volume of ordinary products and new products as well as the optimal proportion of recovered exhaust gas. Moreover, the production cost of new products has a negative impact on all three aspects.

7.2. Theoretical Implications

This paper presents its theoretical significance through the lenses of supply chain, management, and enterprise. Firstly, it introduces a novel approach to studying the traditional supply chain by focusing on the gas supply chain. The analysis highlights that while most studies on waste recovery in the supply chain concentrate on government subsidies and technology inputs, the proportion of recovered waste gas offers a fresh perspective on the development of the gas supply chain and the primary business of the gas company.

Secondly, product quality is guaranteed by the gas concentration (conversion) ratio during the process of remanufacturing and recycling, which provides reference suggestions for enterprises to grasp product quality better and ensure the benefits of technology. Unlike the conventional methods of defining remanufactured products with low recycling quality, this study shows how product quality can be improved and benefits guaranteed from the perspective of gas recovery and conversion.

Finally, based on the waste recycling and reuse model, the influence of various model factors on decision-making without considering the carbon quota and the carbon quota was studied and expanded the research content in the management field. Most of the existing literature uses policy as the influencing factor when exploring the influence on the model of efficiency and profit [41]. Compared to the previous studies, the current research pays more attention to the impact of specific factors in the research process than the impact of policies.

7.3. Managerial Implications

The study has several managerial implications. Firstly, the proportion of waste gas recovered is a new entry point for startups and spinoffs. The study sheds light on the changes brought about by new products entering the market, as well as providing strategies for managing potential risks by sub-dividing the cost of gas containing metal elements. This focus coincides with Yilmaz Bayar et al. and their use of the rate of recovery to deal with optimal decision-making for the recovery, treatment, and reuse of waste materials [22]. Companies can thus ensure product quality by starting with technology that recovers the proportion of the waste gas.

Secondly, the government can adjust its related policies according to actual needs to ensure the interests of all parties and achieve a tripartite environmental–social–economic ecosystem. This is particularly important in the current epidemic, where it is crucial to consider the positive balance of all stakeholders. For example, the Suzhou Municipal Government has taken the lead in supporting small and medium-sized enterprises through its "ten policy opinions" during the business risk period [64]. Producers and recyclers can also achieve a win-win situation, but the government must implement policies in a reasonable manner to attract more participants.

Lastly, recyclers play multiple roles, sharing the pressure for recycling and reusing waste. In this paper, the recycler recycles waste gas, acts as the dealer to sell new products, and even undertakes other manufacturers' work to refine gas. According to solid law, reliable waste treatment only needs to meet the relevant environmental protection requirements and national control standards [65], thus, the task undertaken by the recycler here is exceptionally significant for ecological sustainability.

7.4. Limitation of Research

This paper's limitations suggest one direction for future research. First of all, the model only studied a single period due to the limit of the recovery stage, and there is a lot of space still to search for two periods, multi-periods, and an infinite period. Secondly, more main bodies related to processing and transportation can be designed to play games, but only a producer and a recycler exist. Finally, the Stackelberg model was used to study this model, but other models have not yet had a thorough inquiry, i.e., the Nash equilibrium, and these could also offer significant additional research on the topic discussed here. Not only that, this paper is limited by the carbon quota, but some future changes in laws and policies will create a different environment for research, which is a point of entry.

Author Contributions: K.W. and X.L. were responsible for the research design and conception. H.L. made the bilingual revision of the paper. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Data available on request due to restrictions eg privacy or ethical.

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A

According to the Stackelberg model, the manufacturer is the leader, and the recycler is the follower. Therefore, the proportion of waste gas $\tau_1^*(\tau_2^*)$ recovered by the recycler must be determined first. Secondly, according to the reverse induction method, select the wholesale gas price of the manufacturer $w_1^*(w_2^*)$. Then, determine the number of ordinary products $q_{s1}^*(q_{s2}^*)$ sold by the manufacturer to the market. Finally, the formula $q_r = \beta \cdot \tau \cdot q_s$, $\beta \in (0, 1]$ determines the quantity $q_{r1}^*(q_{r2}^*)$. In addition, the price $p_{s1}^*(p_{s2}^*)$ at which the manufacturer sells ordinary products to the market and the price $p_{r1}^*(p_{r2}^*)$ at which the recycler sells new products.

Appendix A.1. Without Considering the Carbon Quota

Substituting Formula (2) into Formula (6) to obtain:

$$\pi_{r1} = \theta [1 - q_{s1}(\tau_1 + 1)] \cdot q_{s1} \cdot \tau_1 - c_r \cdot \tau_1 \cdot q_{s1} - w_1 \cdot \tau_1 \cdot q_{s1}$$
(A1)

$$\frac{\partial \pi_{r1}}{\partial \tau} = -\theta \cdot \tau_1 \cdot q_{s1}^2 + \theta \cdot [1 - q_{s1} \cdot (\tau_1 + 1)]q_{s1} - c_r \cdot q_{s1} - c_t \cdot q_{s1} - w_1 \cdot q_{s1}$$
(A2)

If $0 < \theta < 1$, $q_s > 0$, it can be seen that $\frac{\partial^2 \pi_{r_1}}{\partial \tau_1^2} = -2q_{s_1}^2 \cdot \theta < 0$, $\theta < 0$, so there is an optimal solution to the profit function of the recycler:

$$-\theta \cdot \tau_1 \cdot q_{s1}^2 + \theta \cdot [1 - q_{s1}(\tau_1 + 1)]q_{s1} - c_r \cdot q_{s1} - c_t \cdot q_{s1} - w_1 \cdot q_{s1} = 0$$
(A3)

Can be obtained by item transfer:

$$q_{s1} = -\frac{c_t + c_r - \theta + w}{\theta(2\tau + 1)} \tag{A4}$$

According to the inverse induction method, replace Formula (3) with Formula (4) to obtain:

$$\pi_{m1} = -\frac{w_1(q_{s1} \cdot \theta + c_r + \delta \cdot c_t - \theta + w_1)}{2\theta} + [1 - q_{s1}(-\frac{q_{s1} \cdot \theta + c_r + \delta \cdot c_t - \theta + w_1}{2q_{s1}} + 1) - c_s]q_{s1}$$
(A5)

From Equation (A1), we can obtain that the Hessian matrix of the profit function of the manufacturer is

$$\begin{bmatrix} -\frac{1}{\theta} & 0\\ 0 & \theta - 2 \end{bmatrix}$$
(A6)

If $D_1 = -\frac{1}{\theta} < 0$, $D_2 = -\frac{1}{\theta}(\theta - 2) > 0$, the Hessian matrix is negative definite, and the profit function of the manufacturer is concave, so there is an optimal solution:

$$-\frac{q_{s1}\cdot\theta+c_r+\delta\cdot c_t-\theta+w_1}{2\theta}-\frac{w}{2\theta}+\frac{q_{s1}}{2}=0$$
(A7)

$$-\frac{w_{1}}{2} + \left[\frac{q_{s1} \cdot \theta + \delta \cdot c_{r} - \theta + w_{1}}{2q_{s1}} - 1 - q_{s1}\left(-\frac{\theta}{2q_{s1}} + \frac{q_{s1} \cdot \theta + c_{r} + \delta \cdot c_{t} - \theta + w_{1}}{2q_{s1}^{2}}\right)\right]q_{s1} + 1 - q_{s1}\left(-\frac{q_{s1} \cdot \theta + c_{r} + \delta \cdot c_{t} - \theta + w_{1}}{2q_{s1}^{2}} + 1\right) - c_{s} = 0 \quad (A8)$$

Solve Equations (A3), (A7) and (A8) to obtain:

$$\tau_1^* = \frac{[(\delta - 1)\theta - 4\delta + 2]c_t + [(1 - \delta)\theta + 2\delta - 4]c_r + 2\theta c_s}{2\theta(\delta c_t + c_r - 2c_s - \theta + 2)}$$
(A9)

$$w_1^* = -\frac{\delta c_r}{2} - \frac{c_t}{2} + \frac{\theta}{2}$$
(A10)

$$J_{s1}^{*} = -\frac{c_r - 2c_s + \delta c_t - \theta + 2}{2(\theta - 2)}$$
(A11)

Therefore, according to Formula (1):

$$q_{r1}^{*} = \frac{[(1-\delta)\theta + 4\delta - 2]c_{t} + [(\delta - 1)\theta - 2\delta + 4]c_{r} - 2c_{s}}{4\theta(\theta - 2)}$$
(A12)

Corollary 1 proved: w_1^* is a monotonic subtractive function of c_t .

Corollary 2 proved: q_{s1}^* is the monotonic decreasing function of c_t , and q_{r1}^* is the monotonic increasing function of c_t .

Corollary 3 proved: q_{s1} is a monotonic subtractive function of w_1 .

Corollary 4 proved: From the value of w_1^* , $-\delta c_r - c_t + \theta > 0$. When $\delta > 0$, $c_r < \frac{\theta - c_t}{\delta}$, When $\delta < 0$, $c_r > \frac{\theta - c_t}{\delta}$. Similarly, from the value of q_{s1}^* , $c_r > 2c_s - \delta \cdot c_t + \theta - 2$. Because $2c_s - \delta \cdot c_t + \theta - 2$ and $(\frac{\theta - c_t}{\delta})$, one of them may be greater than zero or less than zero.

So, Proposition 1 and Proposition 2 are proven.

Corollary 5 proved: τ_1^* is the monotonic increasing function of c_t .

Corollary 6 proved: q_{s1} is a monotonic subtractive function of τ_1 , and Formula (1) $q_{r1} = \tau_1 \cdot q_{s1}$ shows that q_{r1} is the monotonic increasing function of τ_1 .

So, Proposition 3 is proven.

Substituting Equations (A9) and (A11) into Formulas (2) and (3) to obtain:

$$p_{s1}^* = \frac{(\delta - 1)c_t + (1 - \delta)c_r + 2(c_s + 1)}{4}$$
(A13)

$$p_{r1}^{*} = \frac{[(3\delta - 1)\theta - 4\delta + 2]c_{t} + [(3 - \delta)\theta + 2\delta - 4]c_{r} - 2\theta c_{s} + 2\theta^{2} - 4\theta}{4\theta - 8}$$
(A14)

Corollary 7 proved: p_{s1}^* combined with (A11), it can be seen that p_{s1}^* and q_{s1}^* are monotonically increasing functions of c_s . From the combination of p_{r1}^* and (A12), it can be seen that p_{r1}^* and q_{r1}^* are monotonic subtractive functions of c_s .

So, Proposition 4 is proven.

Substituting Equations (A9)–(A11) into Equations (A1) and (A6):

$$\pi_{m1}^{*} = \frac{1}{\delta\theta(\theta-2)} \left\{ (2-4c_{s})\theta^{2} + [-(c_{r}-c_{t})^{2}\delta^{2} + (2c_{r}^{2}-4c_{r}\cdot c_{t}+4c_{s}\cdot c_{t}+2c_{t}^{2})\delta - c_{r}^{2} + (4c_{s}+2c_{t})c_{r} - c_{t}^{2} - 4(c_{r}-c_{t})^{2}]\theta + 2[(c_{r}-2c_{t})\theta - 2c_{r} + c_{t}](c_{r}\theta + c_{t}) \right\}$$

$$(A15)$$

$$\pi_{r1}^{*} = \frac{\left\{\theta[(c_r - c_t)\delta - c_r - 2c_s + c_t] + (4c_t - 2c_r)\delta + 4c_r - 2c_t\right\}^2}{16\theta(\theta - 2)^2}$$
(A16)

Appendix A.2. Considering the Carbon Quota

Substituting Formula (2) into Formula (7) gives:

$$\pi_{r2} = -\beta \cdot c_t \cdot \tau_2 \cdot \delta \cdot q_{s2} + \theta [1 - q_{s2}(\tau_2 + 1)] \beta \cdot q_{s2} \cdot \tau_2 - \beta \cdot c_r \cdot \tau_2 \cdot q_{s2} - \beta \cdot w_2 \cdot \tau_2 \cdot q_{s2}$$
(A17)

$$\frac{\partial \pi_{r2}}{\partial \tau} = -\beta \cdot c_t \cdot \delta \cdot q_{s2} - \theta \cdot \beta \cdot \tau_2 \cdot q_{s2}^2 + \theta [1 - q_{s2}(\tau_2 + 1)]\beta \cdot q_{s2} - \beta \cdot c_r \cdot q_{s2} - \beta \cdot w_2 \cdot q_{s2}$$
(A18)

If $0 < \theta < 1$, and $q_{s2} > 0$, $\frac{\partial^2 \pi_{r2}}{\partial \tau^2} = -2 \cdot \beta \cdot q_{s2}^2 \cdot \theta < 0$. Therefore, there is an optimal solution to the profit function of the recycler

$$-\beta \cdot c_t \cdot \delta \cdot q_{s2} - \theta \cdot \beta \cdot \tau_2 \cdot q_{s2}^2 + \theta [1 - q_{s2}(\tau_2 + 1)]\beta \cdot q_{s2} - \beta \cdot c_r \cdot q_{s2} - \beta \cdot w_2 \cdot q_{s2} = 0$$
(A19)

According to the inverse induction method, we can also have Equations (A5), (A6), (A7) and (A8) to solve Equations (A7), (A8) and (A19), for obtaining:

$$\tau_2^* = \frac{\theta c_s - \delta c_t - c_r}{\theta (\delta c_t + c_r - 2c_s - \theta + 2)}$$
(A20)

$$w_2^* = -\frac{\delta c_r}{2} - \frac{c_t}{2} + \frac{\theta}{2}$$
 (A21)

$$q_{s2}^{*} = -\frac{c_r - 2c_s + \delta c_t - \theta + 2}{2(\theta - 2)}$$
(A22)

Therefore, according to Formula (1):

$$q_{r2}^{*} = \frac{\beta(-\theta c_s + \delta c_t + c_r)}{4\theta(\theta - 2)}$$
(A23)

Substituting Equations (A20) and (A22) into Formulas (2) and (3) to obtain:

$$p_{s2}^{*} = \frac{c_s + 1}{2} \tag{A24}$$

$$p_{r2}^{*} = \frac{\theta^{2} + \theta(\delta \cdot c_{t} + c_{r} - c_{s} - 2) - \delta \cdot c_{t} - c_{r}}{2\theta - 4}$$
(A25)

Substituting Equations (A20)–(A22) into Equations (A17) and (A5):

$$\pi_{m2}^{*} = \frac{\theta^2 (1 - 2c_s) + \theta [-2 - 2c_s^{*} + (2\delta \cdot c_t + 2c_r + 4)c_s] - (\delta \cdot c_t + c_r)^2}{4\theta (\theta - 2)}$$
(A26)

$$\pi_{r2}^{*} = \frac{\beta(-\theta \cdot c_s + \delta \cdot c_t + c_r)^2}{4\theta(\theta - 2)^2}$$
(A27)

Corollary 8 proved: (1) From the values of w_1^* , w_2^* and q_{s1}^* , q_{s2}^* . (2) Obtained in the same way as corollary 1, corollary 2, corollary 3, corollary 6, and corollary 7, which is omitted.

Corollary 9 proved: (1)
$$p_{s1}^* - p_{s2}^* = p_{r1}^* - p_{r2}^* = -\frac{(1-\delta)(c_r - c_t)}{4}$$
; (2) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (2) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (2) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (3) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (4) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (5) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (7) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (7) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (7) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (7) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (7) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (7) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (7) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (7) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (7) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (7) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (7) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (8) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (9) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (9) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (9) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (9) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (9) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (9) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (9) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (9) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (9) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (9) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (9) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (9) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (9) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (9) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (9) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (9) $\pi_{m1}^* - \pi_{m2}^* = -\frac{(\delta-1)^2(c_r - c_t)^2}{4}$; (9) $\pi_{m1}^$

Corollary 10 proved: From the values of τ_2^* , $c_t(c_r)$ is monotonically increasing function of τ_2^* .

So, Proposition 5 is proven.

References

- Recycling Mode of Foreign Renewable Resources. Available online: https://zhuanlan.zhihu.com/p/452220440 (accessed on 6 September 2022).
- The Open-Air Burning of Waste Hardware in Qingcheng Is Worrying, and the Air Is Filled with Highly Toxic and Carcinogenic Substances. Available online: https://news.sina.com.cn/o/2006-08-26/10339854164s.shtml (accessed on 1 October 2022).
- EU Issues Four Draft Regulations on Sustainable Products. Available online: http://www.clii.com.cn/cyaq/202205/t20220524_ 3953983.html (accessed on 6 October 2022).
- 4. Carry Out Pilot Carbon Emission Trading in Many Places to Help China's Ecological Civilization. Available online: https://baijiahao.baidu.com/s?id=1707341404271055011&wfr=spider&for=pc (accessed on 11 October 2022).
- Can't the Carbon Quota Be Paid, and the Fines Be Paid? Don't Be Naive! Available online: https://baijiahao.baidu.com/s?id=17 28524303135257038&wfr=spider&for=pc (accessed on 9 October 2022).
- In 2022, the Global Waste Index Was Released, and the High Recovery Rate of Waste Came from Energy 'Recovery' Rather than Secondary Utilization. Available online: https://www.xianjichina.com/special/detail_507365.html (accessed on 6 September 2022).
- Carbon Trading + Carbon Emission, Six Knowledge Points for the Solid Waste Industry. Available online: https://zhuanlan. zhihu.com/p/382938662 (accessed on 1 October 2022).

- Jinhong Gas Is Better than Huate Gas, and Gas Is Called 'The Blood of Industry'. Available online: https://guba.eastmoney.com/ news,688106,939066180.html (accessed on 14 August 2022).
- Jinhong Gas Has Been Investigated by 12 Institutions: The Company's Ultra Pure Ammonia and High-Purity Laughing Gas Have a High Domestic Market Share. Available online: http://stock.10jqka.com.cn/20211222/c635323548.shtml (accessed on 14 August 2022).
- Miranda, I.T.P.; Fidelis, R.; de Souza Fidelis, D.A.; Pilatti, L.A.; Picinin, C.T. The integration of recycling cooperatives in the formal management of municipal solid waste as a strategy for the circular economy—The case of Londrina, Brazil. *Sustainability* 2020, 12, 10513. [CrossRef]
- 11. Sustainable Development: Germany's Largest Retail Giants Have Successively Transformed into Waste Recycling and Treatment Companies. Available online: https://www.shangyexinzhi.com/article/4251939.html (accessed on 7 September 2022).
- The Biggest Cooperation Commitment between Dow and Mura So Far Is to Promote Advanced Plastic Recycling Business on a Large Scale. Available online: https://baijiahao.baidu.com/s?id=1739021912230387861&wfr=spider&for=pc (accessed on 7 September 2022).
- Xing, E.; Shi, C.; Yan, X.; Cheng, S.; Lin, J.; Ni, S. Research on Collaborative low-carbon emission reduction of enterprises under three-dimensional trading mode. *Chin. J. Manag. Sci.* 2020, 28, 8. Available online: https://kns.cnki.net/kcms/detail/detail. aspx?FileName=ZGGK202003018&DbName=DKFX2020 (accessed on 6 August 2022).
- Huang, Z.; Jiang, L.; Meng, L. Research on optimal decision mechanism considering consumer preference under supply chain competition. *Ind. Eng. Manag.* 2020, 25, 9. Available online: https://kns.cnki.net/kcms/detail/detail.aspx?FileName=GYGC202 006009&DbName=DKFX2020 (accessed on 6 August 2022).
- Liang, W.; Lai, J. Research on green supply chain decision making considering flexible options and forward contracts. *Chin. J. Manag. Sci.* 2022, *13*, 1. Available online: https://kns.cnki.net/kcms/detail/detail.aspx?FileName=ZGGK20220316008 &DbName=DKFX2022 (accessed on 6 August 2022).
- 16. Vrabie, C. Converting municipal waste to energy through the biomass chain, a key technology for environmental issues in (Smart) cities. *Sustainability* **2021**, *13*, 4633. [CrossRef]
- Li, J.; Liu, G.; Zhang, J. Decentralized intrusion strategy of manufacturers based on consumers' green preference and channel competition. *Chinese J. Manag. Sci.* 2022, *8*, 6. Available online: https://kns.cnki.net/kcms/detail/detail.aspx?FileName=ZGGK2 0220516000&DbName=DKFX2022 (accessed on 6 August 2022).
- Sang, S.; Zhang, Q. Research on optimal decision of green supply chain based on Uncertainty Theory. *Chinese J. Manag. Sci.* 2020, 28, 10. Available online: https://kns.cnki.net/kcms/detail/detail.aspx?FileName=ZGGK202009013&DbName=DKFX2020 (accessed on 6 August 2022).
- 19. Suzhou Moore Gas Co., Ltd. Available online: https://www.moorgas.cn/ (accessed on 9 September 2022).
- Lou, G.; Qiu, Y.; Xia, H. Green supply chain coordination considering R & D efforts and marketing investment. *Ind. Eng. Manag.* 2020, 25, 9. Available online: https://kns.cnki.net/kcms/detail/detail.aspx?FileName=GYGC202004016&DbName=DKFX2020 (accessed on 6 August 2022).
- Kim, J.; Park, S.; Kim, H.M. Optimal Modular Remanufactured Product Configuration and Harvesting Planning for End-of-Life Products. J. Mech. Des. 2022, 4, 144. Available online: https://xueshu.baidu.com/usercenter/paper/show?paperid=1k0g0r00rk2 m0ga0fy3c08q057283091&site=xueshu_se (accessed on 6 February 2023). [CrossRef]
- 22. Bayar, Y.; Gavriletea, M.D.; Sauer, S.; Paun, D. Impact of municipal waste recycling and renewable energy consumption on co2 emissions across the european union (Eu) member countries. *Sustainability* **2021**, *13*, 656. [CrossRef]
- Cao, X.; Wu, H.; Wen, H. Emergency and Coordination Decision of Closed loop Supply Chain under the Disturbance of Advertising Effect. J. Manag. Sci. China 2022, 3, 25. Available online: https://kns.cnki.net/kcms/detail/detail.aspx?dbcode= CJFD&dbname=CJFDLAST2022&filename=JCYJ202203006&uniplatform=NZKPT&v=SZEcVr42kJZzch9IyZkz1fJae9FANIX3 3m8UUEwCClkqOoQlV0cWLtyh9y7cQCbC (accessed on 14 September 2022).
- 24. Zhao, Y.; Wang, W.; Ni, Y. EPR system based on a reward and punishment mechanism: Producer-led product recycling channels. *Waste Manag.* **2020**, *103*, 198–207. [CrossRef] [PubMed]
- Huang, H.; Zhou, X.; Zhang, J. Pricing of two-way two-channel closed-loop supply chain of new energy vehicles under government subsidies. *Syst. Eng.* 2020, *38*, 9. Available online: https://kns.cnki.net/kcms/detail/detail.aspx?FileName=GCXT2 02004007&DbName=DKFX2020 (accessed on 6 September 2022).
- Tan, M.; Wang, B.; Zheng, K.; Cheng, H. Pricing Strategies of Dual-Recycling Channels considering Refurbishing and Remanufacturing of WEEE. *Math. Probl. Eng.* 2022, 2022, A9000057. Available online: https://xueshu.baidu.com/usercenter/paper/show? paperid=1j3s0cb0q0210010e52002c0n7416412&site=xueshu_se (accessed on 14 September 2022). [CrossRef]
- Su, L.; Lei, Y.; Junyuan, Q.; Yao, W. Pricing Strategy and Recycling Mode Selection of Closed loop Supply Chain in the Presence of C2C Second hand Market. *Chin. J. Manag.* 2022, *4*, 19. Available online: https://kns.cnki.net/kcms/detail/detail.aspx?dbcode= CJFD&dbname=CJFDLAST2022&filename=GLXB202204012&uniplatform=NZKPT&v=dbPcAeDTFL7GY3b0RTdXsNjho8 3TieQRXc_5oKPYZwKDcYkr47fUSOtYUGYETbVp (accessed on 14 September 2022).
- Li, S.; Chen, D. Selection and coordination of manufacturer's recycling partners in closed-loop supply chain considering CSR input. J. Ind. Eng. Eng. Manag. 2021, 6, 35. Available online: https://kns.cnki.net/kcms/detail/detail.aspx?dbcode=CJFD& dbname=CJFDLAST2021&filename=GLGU202106014&uniplatform=NZKPT&v=rcYoFeiz3PeT_jA9T_MkI65s4Ymye4YRQHT_ YOzt2Ipr1adhi7ppar_ToMSIrYcL (accessed on 14 September 2022).

- 29. Saja, A.M.A.; Zimar, A.M.Z.; Junaideen, S.M. Municipal solid waste management practices and challenges in the southeastern coastal cities of Sri Lanka. *Sustainability* **2021**, *13*, 4556. [CrossRef]
- Huang, X.; Yang, S.; Shi, X. How corporate social responsibility and external stakeholder concerns affect green supply chain cooperation among manufacturers: An interpretive structural modeling analysis. *Sustainability* 2021, 13, 2518. [CrossRef]
- Orlov, A.; Klyuchnikova, E.; Korppoo, A. Economic and environmental benefits from municipal solid waste recycling in the Murmansk region. *Sustainability* 2021, 13, 10927. [CrossRef]
- 32. Pfadt-Trilling, A.R.; Volk, T.A.; Fortier, M.O.P. Climate Change Impacts of Electricity Generated at a Waste-to-Energy Facility. *Environ. Sci. Technol.* **2021**, *55*, 3. [CrossRef]
- 33. Allam, S.Z. De-carbonized energy initiative with bio-cell-distributed stations using GIS geodesic tools towards circular economy. Energy Environ. 2022, 33, 3. Available online: https://xueshu.baidu.com/usercenter/paper/show?paperid=1y1406p05x360e0 0nm7s0tx0u0155357&site=xueshu_se (accessed on 6 February 2023). [CrossRef]
- Peng, B.; Yu, T. Research on electronic waste recycling from the perspective of social capital embedding. J. Ind. Eng. Eng. Manag. 2020, 2, 34. Available online: https://kns.cnki.net/kcms/detail/detail.aspx?dbcode=CJFD&dbname=CJFDLAST202 0&filename=GLGU202002012&uniplatform=NZKPT&v=SJgozF9TV5zgvv9gUewQXTmY9jkjgyQA39HCpMmdJ99Oer9-zHb6 SBvpSugcnCn8 (accessed on 14 September 2022).
- 35. Zhu, X.; Wei, L. Research on two-stage supply chain recovery pricing model considering consumer behavior in bilateral network environment. *Chinese J. Manag. Sci.* 2021, 5, 97–107. Available online: https://kns.cnki.net/kcms/detail/detail.aspx?dbcode= CJFD&dbname=CJFDLAST2021&filename=ZGGK202105010&uniplatform=NZKPT&v=bJQAXkoopAibOYtQ-sxqidRm-_rK7 VrM78I7rg2TF2Y9F84ZIJrK3-iXWQjlKY8W (accessed on 9 September 2022).
- Kwak, M. Monte carlo simulation of the effect of heterogeneous too-cheap prices on the average price preference for remanufactured products. *Sustainability* 2021, 13, 9498. [CrossRef]
- Mustajib, M.I.; Ciptomulyono, U.; Kurniati, N. A novel multi-criteria sorting model based on ahp-entropy grey clustering for dealing with uncertain incoming core quality in remanufacturing systems. *Appl. Sci.* 2021, 11, 2731. [CrossRef]
- Xu, M.; Wang, J.; Jian, H. Pricing and coordination of closed-loop supply chain under uncertainty of patent protection and output. J. Ind. Eng. Eng. Manag. 2021, 35, 11. Available online: https://kns.cnki.net/kcms/detail/detail.aspx?FileName=GLGU2021030 12&DbName=CJFQ2021 (accessed on 6 September 2022).
- 39. Zheng, B.; Yu, N.; Chu, J. Managing Sales Channel Selection for a Manufacturer in the Presence of Remanufacturing. *J. Syst. Sci. Syst. Eng.* **2021**, *30*, 600–625. [CrossRef]
- Bai, Q.; Xu, J.; Zhang, Y. The distributionally robust optimization model for a remanufacturing system under cap-and-trade policy: A newsvendor approach. *Ann. Oper. Res.* 2022, *2*, 731–760. Available online: https://xueshu.baidu.com/usercenter/ paper/show?paperid=1k770a00am4c0rt0971u0cv0my226059&site=xueshu_se (accessed on 15 September 2022). [CrossRef]
- 41. Khan, S.A.R.; Godil, D.I.; Thomas, G.; Tanveer, M.; Zia-Ul-haq, H.M.; Mahmood, H. The decision-making analysis on end-of-life vehicle recycling and remanufacturing under extended producer responsibility policy. *Sustainability* **2021**, *13*, 11215. [CrossRef]
- Zhang, L.; Zifan, Y.; Chunqi, C. Selection of emission reduction strategies for closed-loop supply chain under carbon quota trading policy. J. Ind. Eng. Eng. Manag. 2022, 1, 36. Available online: https://kns.cnki.net/kcms/detail/detail.aspx?dbcode=CJFD& dbname=CJFDLAST2022&filename=GLGU202201015&uniplatform=NZKPT&v=_6D3yca4QpMm0iopQ7D6tWSFKwO66-3HyQTd6PxgFrGnezS7wJ7h9662Cy-fj0-i (accessed on 16 September 2022).
- 43. Meng, L.J.; Huang, Z.Q.; Zhang, B.Y.; Yang, Y.X. Analysis on the Policy Model of Consumption Subsidies for Remanufactures in Closed loop Supply Chain. *Chin. J. Manag. Sci.* 2021, *8*, 29. Available online: https://kns.cnki.net/kcms/detail/ detail.aspx?dbcode=CJFD&dbname=CJFDLAST2021&filename=ZGGK202108014&uniplatform=NZKPT&v=bJQAXkoopAg2 qQyKIcwuXHDAjBL8dESnalArw1pKClj4OboEOMGKXgNRE8ApS04E (accessed on 16 September 2022).
- 44. Han, T.; Li, L.; Jing, H. Research on decision-making of dual channel green supply chain considering government subsidies and fairness concerns. *Chin. J. Manag. Sci.* 2022, 4, 30. Available online: https://kns.cnki.net/kcms/detail/detail.aspx? dbcode=CJFD&dbname=CJFDLAST2022&filename=ZGGK202204017&uniplatform=NZKPT&v=Nd7LtefyLHgXss8CZsX7 AXeUcwZVSK2p1ixOGWgzlr-9apZKnFwJvg8-Q-iRhdre (accessed on 16 September 2022).
- 45. Xia, X.; Biao, L. Research on the impact of government carbon tax and subsidy policies on outsourcing remanufacturing. *Chin. J. Manag. Sci.* 2022, 9, 30. Available online: https://kns.cnki.net/kcms/detail/detail.aspx?dbcode=CJFD&dbname=CJFDAUTO&filename=ZGGK202209010&uniplatform=NZKPT&v=Nd7LtefyLHieWDCd2KhYSO4EfvtaNye6R0sTvQ_DOrr-OQ_080BX3 MZkBOe0DoFY (accessed on 16 September 2022).
- 46. Mead, N.L.; Williams, L.E. The Pursuit of Meaning and the Preference for Less Expensive Options. *J. Consum. Res.* 2023, 49, 5. [CrossRef]
- 47. Attaran, M. 3D Printing Role in Filling the Critical Gap in the Medical Supply Chain during COVID-19 Pandemic. *Am. J. Ind. Bus. Manag.* 2020, *10*, 05. [CrossRef]
- Ling Hu, D.; Wu, X.; Ye, F. Research on carbon quota allocation and trading mechanism considering double heterogeneity. *Chinese J. Manag. Sci.* 2021, 29, 12. Available online: https://kns.cnki.net/kcms/detail/detail.aspx?FileName=ZGGK202103017 &DbName=DKFX2021 (accessed on 6 August 2022).
- Tan, Q.; Zheng, J.; Ding, Y.; Zhang, Y. Provincial carbon emission quota allocation study in china from the perspective of abatement cost and regional cooperation. *Sustainability* 2020, 12, 8457. [CrossRef]

- Liu, C.; Tang, Z. Integrated optimization of sustainable level of inventory transportation under carbon quota policy. *Ind. Eng.* 2022, 25, 10. Available online: https://kns.cnki.net/kcms/detail/detail.aspx?FileName=GDJX202203008&DbName=CJFQTEMP (accessed on 6 August 2022).
- 51. Guo, J.; Sun, L.; Zhang, C.; Ni, M.; Zhu, J. Pricing and coordination of supply chain considering low carbon preference of consumers under carbon cap and trade policy. *Ind. Eng. Manag.* 2020, 25, 12. [CrossRef]
- 52. Zhang, Y.; Tan, D. Research on the influence of double carbon policy on the decision-making of manufacturing enterprises and the government regulation strategy. *Manage. Rev.* 2022, 34, 12. [CrossRef]
- 53. Shi J., Z.; Wenxin, J.; Qiang, D.; Jianqiang, L. Research on supply chain contract coordination with capital constraints under two-way carbon emission reduction cost sharing. *Syst. Sci. Math.* 2023, *1*, 13. Available online: https://kns.cnki.net/kcms2 /article/abstract?v=3uoqIhG8C45S0n9fL2suRadTyEVl2pW9UrhTDCdPD67ID9CfkQCBqb6Qsan06h_YirWYmQDcb6w7UIr3 AumkITZ6of7yXhTK&uniplatform=NZKPT (accessed on 6 February 2023).
- Xu, G.; Yue, D. Supply chain pricing decision of one manufacturer and two retailers under carbon quota policy. *Syst. Sci. Math.* 2020, 40, 16. Available online: https://kns.cnki.net/kcms/detail/detail.aspx?FileName=STYS202006005&DbName=DKFX2020 (accessed on 6 August 2022).
- 55. Zhang, H.; Aydin, G.; Parker, R.P. Social Responsibility Auditing in Supply Chain Networks. *Manage. Sci.* 2022, 68, 1058–1077. [CrossRef]
- 56. Karimi, M.; Zaerpour, N. Put your money where your forecast is: Supply chain collaborative forecasting with cost-function-based prediction markets. *Eur. J. Oper. Res.* 2022, 300, 1035–1049. Available online: https://xueshu.baidu.com/usercenter/paper/show?paperid=1h4n0ca0rx5n0c20t76t0220re366725&site=xueshu_se (accessed on 24 January 2023). [CrossRef]
- 57. Hallman, N.J.; Kartapanis, A.; Schmidt, J.J. How do auditors respond to competition? Evidence from the bidding process. J. Account. Econ. 2022, 73, 101475. Available online: https://xueshu.baidu.com/usercenter/paper/show?paperid=1y2208p0ub1y0 c80v54q0440ek185481&site=xueshu_se (accessed on 24 January 2023). [CrossRef]
- 58. Zhang, W.; Wu, D.; Su, C.; Yang, L.; Liu, L. Information Bottleneck and Breakthrough in Solid Waste Management: Construction and Application of Risk-Based Solid Waste Environmental Audit Model. *Res. Sci. Technol. Manag.* 2022, 42, 190–195. Available online: https://kns.cnki.net/kcms/detail/detail.aspx?dbcode=CJFD&dbname=CJFDLAST2022&filename=KJGL2022070 23&uniplatform=NZKPT&v=vw4ppLnL3OYfmagn_fbl6H-0dSCad8FJE2JB6Z32u3dsvyfMyucEyHYw8lWvA_nj (accessed on 24 January 2023).
- Guo, P. Research on audit supervision of agricultural surface source pollution prevention and control. *Environ. Prot.* 2020, 48, 25–29. Available online: https://kns.cnki.net/kcms/detail/detail.aspx?dbcode=CJFD&dbname=CJFDLAST2020&filename=HJBU2 02008010&uniplatform=NZKPT&v=T7mNtCevx2nePy2bUjFFTFEYF0sjdzEjjKW69xx0wR6ipzh8kbHRU8fy71qjLpib (accessed on 24 January 2023).
- 60. What Are the Control Measures for Solid Wastes? Available online: https://www.sohu.com/a/482347587_120797185 (accessed on 11 August 2022).
- 61. Multi Perspective Observation on the Waste Incineration Industry: Changes and Changes in the Post Subsidy Era. Available online: http://www.chinapower.com.cn/jnhb/ljfd/20220606/152282.html (accessed on 11 August 2022).
- Jinfei Kaida: Most of the Company's Products Adopt the Strategy of Linkage between Sales Price and Aluminum Price, and Have Certain Price Transmission and Transfer Ability. Available online: http://www.nbd.com.cn/articles/2021-12-17/2046202.html (accessed on 11 August 2022).
- How Can Share Purchase Empower Enterprises and Merchants? Analysis on the Core Value of the Business Model of Share Purchase. Available online: https://baijiahao.baidu.com/s?id=1740829035110961435&wfr=spider&for=pc (accessed on 11 August 2022).
- 64. Supply Chain Management Risk under Epidemic Situation and Local Government's Precise Measures. Available online: http://finance.sina.com.cn/roll/2020-02-05/doc-iimxxste9130145.shtml (accessed on 12 August 2022).
- 65. Law of the People's Republic of China on the Prevention and Control of Solid Waste Pollution. Available online: https://www.mee.gov.cn/ywgz/fgbz/fl/202004/t20200430_777580.shtml (accessed on 12 August 2022).

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.